Author’s response to reviews

Title: Searching for chronic hepatitis B patients in a low prevalence area - role of racial origin.

Authors:

Dr Suzane K Ono-Nita (skon@usp.br)
Dr Flair J Carrilho (fjcarril@usp.br)
Rita A Cardoso (rita@statistika.com.br)
Marcelo E Nita (marcelo_nita@uol.com.br)
Luiz C Da Silva (lucadasi@bol.com.br)

Version: 6 Date: 23 Mar 2004

Cover letter

Re: Searching for chronic hepatitis B patients in a low prevalence area - role of racial origin.
1180107372542836

Dear Dr. Stokes,

Thank you very much for accepting our manuscript for publication. We are glad to contribute to the journal and hope it will help to the better understanding and improve the hepatitis B patient and family care. Also we would like to express our gratitude to the reviewers for their effort to improve our manuscript with their suggestions.

As requested I made the changes as below:

1. Covering letter and an earlier version of manuscript as additional files were removed.

2. Concerning the use of the term "study" rather than "trial" in the abstract section, please note that it was changed from the old version in the first line from:
"Clinical trials for testing new drugs ..."
to
"Clinical studies for testing new drugs..."

3. Figure 5 was modified as "portrait" as requested.

4. All unnecessary capitalisation from the title was removed

5. I cropped the figures, but since it is the first time I did it, I would be grateful if you could check if it is correct the way it was done.

We hope it is everything correct and would like to thank you again for your kind consideration.

Best regards,

Suzane